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LHC Status (past 2 months)

‣ Focus on delivering colliding beams for physics in August

‣ Instantaneous luminosity peaking at 7.5x1033 cm-2 s-1

‣ Octupole polarity changed, allowing increase in bunch intensities
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        CMS Peak Inst. Lumi (per Fill)‣ In September:

‣ Special physics runs

‣ β* ≥ 500 m run

‣ Pilot proton-lead 
run

‣ Technical stop

‣ a kicker magnet 
at p8 was 
replaced

‣ Physics data again 
over the weekend
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pPb event (13 Sep 2012)
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‣ 2.3M pPb events collected in 4.5 hours
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CMS Activities

‣ Sample of activities over the last 2 months:

‣ Implementation of automatic turn-on of high voltage for tracker 
systems when beam is stable

‣ Planned maintenance on the cooling systems for the tracker

‣ Calorimeter readout tests

‣ L1 tests focussed on reducing the resync and recovery times

‣ Ongoing trigger development to accommodate luminosities near 
8E33

‣ DAQ and router updates
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Summary of Fills
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Efficiency
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Downtimes (past 2 months)

‣ Friday August 10th: cooling failure, solenoid down for 5 days

‣ machine development days advanced to minimize CMS data loss

‣ accounts for most of INFRAstructure category above

8



All Experimenters' Meeting10/01/12

Integrated luminosity
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Plan

‣ 25 ns scrubbing run this week

‣ testing the kicker vacuum with injections today

‣ scrubbing scheduled to begin on 4 Oct

‣ Machine development next week

10

ML

V2.3

 

Jan Feb    Mar

Wk 1 2 3 4 5 6 7 8 9 10 11 12 13

Mo 2 9 16 23 30 6 13 20 27 5 12 19 26

Tu
We 01-­‐Feb

Th   

Fr
Sa
Su

  

       

Apr May June

Wk 14 15 16 17 18 19 20 21 22 23 24 25 26

Mo 2 9 16 23 30 7 14 21 28 4 11 18 25

Tu
We
Th
Fr
Sa
Su

     

July Aug    Sep

Wk 27 28 29 30 31 32 33 34 35 36 37 38 39

Mo 2 9 16 23 30 6 13 20 27 3 10 17 24

Tu
We
Th
Fr
Sa
Su

  

Oct Nov       Dec

Wk 40 41 42 43 44 45 46 47 48 49 50 51 52

Mo 1 8 15 22 29 5 12 19 26 3 10 17 24

Tu
We
Th   

Fr      

Sa
Su

  

Technical  Stop   

Recommissoning  with  beam      

   Machine  development   

Special  physics  runs

September 14, 2012

    

2012  LHC  Schedule  
Approved  by  the  Reseach  Board,  December  2011.  

Easter 

G. Friday 

1st May 

Whit 

      

Ascension 

Re-­‐commissioning  
with  beam  

J. Genevois 

Technical stop 
Hardware 

commissioning 

MD  

Machine  
checkout  

Start  partial  
powering  tests  

Start  full  powering  
tests  

Xmas 

DSO  tests  

Floating  MD  
[pA]  

MD  

90  m  [12  h]  

Floating  MD  
[48  h]  

90  m  
[24  h]  

VdM  scans  
[48  h]  

90  m  
[24  h]  

VdM  scans  
[48  h]  

MD  4  

TS3  

TS2  TS1  

500+  m  

500+  m  
[24  h]  

MD  3  

Christmas  
technical  stop  

Pilot  pA  run  

  Ions  available  
again  

Scrubbing  
run  

    

Floating  MD  
[24  h]  

ML

V2.3

 

Jan Feb    Mar

Wk 1 2 3 4 5 6 7 8 9 10 11 12 13

Mo 2 9 16 23 30 6 13 20 27 5 12 19 26

Tu
We 01-­‐Feb

Th   

Fr
Sa
Su

  

       

Apr May June

Wk 14 15 16 17 18 19 20 21 22 23 24 25 26

Mo 2 9 16 23 30 7 14 21 28 4 11 18 25

Tu
We
Th
Fr
Sa
Su

     

July Aug    Sep

Wk 27 28 29 30 31 32 33 34 35 36 37 38 39

Mo 2 9 16 23 30 6 13 20 27 3 10 17 24

Tu
We
Th
Fr
Sa
Su

  

Oct Nov       Dec

Wk 40 41 42 43 44 45 46 47 48 49 50 51 52

Mo 1 8 15 22 29 5 12 19 26 3 10 17 24

Tu
We
Th   

Fr      

Sa
Su

  

Technical  Stop   

Recommissoning  with  beam      

   Machine  development   

Special  physics  runs

September 14, 2012

    

2012  LHC  Schedule  
Approved  by  the  Reseach  Board,  December  2011.  

Easter 

G. Friday 

1st May 

Whit 

      

Ascension 

Re-­‐commissioning  
with  beam  

J. Genevois 

Technical stop 
Hardware 

commissioning 

MD  

Machine  
checkout  

Start  partial  
powering  tests  

Start  full  powering  
tests  

Xmas 

DSO  tests  

Floating  MD  
[pA]  

MD  

90  m  [12  h]  

Floating  MD  
[48  h]  

90  m  
[24  h]  

VdM  scans  
[48  h]  

90  m  
[24  h]  

VdM  scans  
[48  h]  

MD  4  

TS3  

TS2  TS1  

500+  m  

500+  m  
[24  h]  

MD  3  

Christmas  
technical  stop  

Pilot  pA  run  

  Ions  available  
again  

Scrubbing  
run  

    

Floating  MD  
[24  h]  


